

# Schistosoma haematobiuminfection is associated with lower serum cholesterol levels and improved lipid profile in overweight/obese individuals

Zinsou, J.F.; Janse, J.J.; Honpkehedji, J.Y.; Dejon-Agobe, J.C.; Garcia-Tardon, N.; Hoekstra, P.T.; ... ; Guigas, B.

## Citation

Zinsou, J. F., Janse, J. J., Honpkehedji, J. Y., Dejon-Agobe, J. C., Garcia-Tardon, N., Hoekstra, P. T., ... Guigas, B. (2020). Schistosoma haematobiuminfection is associated with lower serum cholesterol levels and improved lipid profile in overweight/obese individuals. *Plos Neglected Tropical Diseases*, *14*(7). doi:10.1371/journal.pntd.0008464.r004

Version:Publisher's VersionLicense:Creative Commons CC BY 4.0 licenseDownloaded from:https://hdl.handle.net/1887/3181371

Note: To cite this publication please use the final published version (if applicable).



# GOPEN ACCESS

**Citation:** Zinsou JF, Janse JJ, Honpkehedji YY, Dejon-Agobé JC, García-Tardón N, Hoekstra PT, et al. (2020) *Schistosoma haematobium* infection is associated with lower serum cholesterol levels and improved lipid profile in overweight/obese individuals. PLoS Negl Trop Dis 14(7): e0008464. https://doi.org/10.1371/journal.pntd.0008464

**Editor:** Keke Fairfax, University of Utah, UNITED STATES

Received: March 4, 2020

Accepted: June 8, 2020

Published: July 2, 2020

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0008464

**Copyright:** © 2020 Zinsou et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript and its Supporting Information files.

**RESEARCH ARTICLE** 

# *Schistosoma haematobium* infection is associated with lower serum cholesterol levels and improved lipid profile in overweight/obese individuals

Jeannot F. Zinsou<sup>1,2</sup>, Jacqueline J. Janse<sup>1</sup>, Yabo Y. Honpkehedji<sup>1,2</sup>, Jean Claude Dejon-Agobé<sup>2</sup>, Noemí García-Tardón<sup>1</sup>, Pytsje T. Hoekstra<sup>1</sup>, Marguerite Massinga-Loembe<sup>2,5,6</sup>, Paul L. A. M. Corstjens<sup>3</sup>, Govert J. van Dam<sup>1</sup>, Martin Giera<sup>4</sup>, Peter G. Kremsner<sup>2,5</sup>, Maria Yazdanbakhsh<sup>1</sup>, Ayola A. Adegnika<sup>1,2,5,6</sup>, Bruno Guigas<sup>1</sup>\*

1 Department of Parasitology, Leiden University Medical Center, Leiden, The Netherlands, 2 Centre de Recherches Médicales de Lambaréné, Lambaréné, Gabon, 3 Department of Cell and Chemical Biology, Leiden University Medical Center, Leiden, The Netherlands, 4 Center for Proteomics and Metabolomics, Leiden University Medical Center, Leiden, The Netherlands, 5 Institut für Tropenmedizin, Universität Tübingen, Tübingen, Germany, 6 German Center for Infection Research, Tübingen, Tübingen, Germany

\* b.g.a.guigas@lumc.nl

# Abstract

Infection with parasitic helminths has been reported to improve insulin sensitivity and glucose homeostasis, lowering the risk for type 2 diabetes. However, little is known about its impact on whole-body lipid homeostasis, especially in obese individuals. For this purpose, a crosssectional study was carried out in lean and overweight/obese adults residing in the Lambaréné region of Gabon, an area endemic for Schistosoma haematobium. Helminth infection status, peripheral blood immune cell counts, and serum metabolic and lipid/lipoprotein levels were analyzed. We found that urine S. haematobium egg-positive individuals exhibited lower serum total cholesterol (TC; 4.42 vs 4.01 mmol/L, adjusted mean difference [95%CI] -0.30 [-0.68,-0.06]; P = 0.109), high-density lipoprotein (HDL)-C (1.44 vs 1.12 mmol/L, -0.24 [-0.43,-0.06]; P = 0.009) and triglyceride (TG; 0.93 vs 0.72 mmol/L, -0.20 [-0.39,-0.03]; P = 0.022) levels than egg-negative individuals. However, when stratified according to body mass index, these effects were only observed in overweight/obese infected individuals. Similarly, significant negative correlations between the intensity of infection, assessed by serum circulating anodic antigen (CAA) concentrations, and TC (r = -0.555; P<0.001), HDL-C (r = -0.327; P = 0.068), LDL-C (r = -0.396; P = 0.025) and TG (r = -0.381; P = 0.032) levels were found in overweight/obese individuals but not in lean subjects. Quantitative lipidomic analysis showed that circulating levels of some lipid species associated with cholesterol-rich lipoprotein particles were also significantly reduced in overweight/obese infected individuals in an intensity-dependent manner. In conclusion, we reported that infection with S. haematobium is associated with improved lipid profile in overweight/obese individuals, a feature that might contribute reducing the risk of cardiometabolic diseases in such population.

**Funding:** This work was supported by funding from the Dutch Organization for Scientific Research: [ZonMw TOP Grant 91214131 to MY and BG] & [NWO project 184.034.019 to MG] https://www.zonmw.nl/nl/. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Infection with parasitic helminths has been reported to be beneficial for metabolic homeostasis by improving insulin sensitivity and lowering the risk for developing type 2 diabetes. Elevated circulating cholesterol and triglyceride levels associated with obesity are also risk factors for cardiometabolic diseases. In the framework of a cross-sectional study conducted in an endemic rural area, we have investigated the impact of infection with *Schistosoma hematobium* on serum lipid homeostasis in adult individuals with a broad range of body weight. We found that helminth infection is associated with a lower serum total cholesterol (TC), high-density lipoprotein (HDL)-C and triglyceride (TG) levels, especially in overweight/obese individuals. Furthermore, significant negative correlations between the intensity of infection and TC, HDL-C, LDL-C and TG levels were also found in overweight/obese individuals but not in lean subjects. Altogether our study show for the first time that infection with *Schistosoma hematobium* is associated with an improved serum lipid profile in overweight/obese humans, a feature that may contribute to protection against cardiometabolic diseases in such population. Further investigation is however required to elucidate the underlying molecular mechanisms.

### Introduction

About one quarter of Earth inhabitants are infected with parasitic helminths and most of them are living in Africa [1]. In tropical and subtropical areas, soil-transmitted helminths (STH) are widely distributed, including *Ascaris lumbricoides*, *Trichuris trichiura* and hookworms, together with filarial nematodes and schistosomes such as *Schistosoma mansoni* and *Schistosoma haematobium* [1, 2]. Schistosomiasis affects 200 to 300 million people worldwide [1], especially in the poorest regions of the low and middle income countries (LMIC), and remains a major public health problem in these endemic areas [3, 4], where it co-exists with other health condition, including non-communicable diseases (NCDs).

NCDs such as obesity, metabolic syndrome and type 2 diabetes (T2D) are increasing not only in industrialized countries but also in developing countries that experienced rapid ruralurban transition, notably in Africa [5–7]. In 2017, 16 million African people were estimated to have T2D, with a projected doubling of this number in the coming 20 years [5]. Obesity is often associated with chronic low-grade inflammation, also called meta-inflammation, which contributes to peripheral and systemic insulin resistance, alterations of glucose and lipid metabolism, and ultimately leads to the development of T2D and related cardiovascular diseases [8]. Meta-inflammation mostly results from alterations in the composition and/or activation state of a variety of innate and adaptive immune cells in metabolic organs, notably in adipose tissue and the liver, creating a pro-inflammatory environment that contribute to tissue-specific insulin resistance and whole-body metabolic dysfunctions [9].

Chronic helminth infection is known to trigger a potent  $T_H2$  immune response together with T-cell hyporesponsiveness through induction of a regulatory network [10], which have been both suggested to dampen meta-inflammation and restore tissue-specific and wholebody insulin sensitivity [9]. Interestingly, helminth infection and treatment with helminthderived molecules have also been shown to improve insulin sensitivity and glucose homeostasis in various rodent models of T2D, at least partly through induction of a type 2 immune response in metabolic organs [11]. In line with this, epidemiological studies conducted in countries endemic for various helminth species have reported an inverse association between helminth infection and insulin resistance [12] or incidence of metabolic syndrome and T2D [13, 14], suggesting a positive impact of helminths on host metabolic homeostasis. More recently, clinical interventional trials using anthelmintic drugs have also shown that deworming was associated with impaired systemic insulin sensitivity and glucose homeostasis in either STH-infected individuals or type 2 diabetic subjects with *Strongyloides stercoralis* infection, respectively [15, 16]. Nevertheless, the few human studies performed so far have mostly been done in STH-infected lean individuals, and principally focused on glucose rather than lipid homeostasis [11]. As such, little is currently known on the impact of other helminth infection, notably with schistosomes, on metabolic homeostasis in people with a broad range of body weight. We have therefore conducted a cross-sectional study to investigate whether infection with *S. haematobium* could affect glucose and lipid homeostasis in lean and obese/overweight adult individuals living in endemic rural area.

#### Methods

#### **Ethics statement**

The study was approved by the institutional ethics committee of the Centre de Recherches Médicales de Lambaréné (CERMEL), Lambaréné, Gabon (Registered number: CEI-MRU 002/ 2014) and was conducted in line with the Good Clinical Practice principles of the International Conference on Harmonization and the Declaration of Helsinki.

#### Study site

The study was conducted in Lambaréné region, an endemic area for *S. haematobium* in Gabon. The schistosomiasis prevalence among children and young adults was recently found to be 26% in Lambaréné village (Jean Claude Dejon-Agobé, personal communication) and 45% in the surrounding Zilé-PK area [17, 18].

#### Participants and study procedures

Adult individuals (>18 year-old) with either low/normal (<25) or high (>25) BMI and living in the study area were invited to join the study during on-site visit. An a priori power calculation was performed using mean serum TC levels from a previous small cohort study to determine the sample size allowing to detect a mean difference of  $\sim 12,5\%$  between Sh<sup>-</sup> and Sh<sup>+</sup> group ( $\alpha = 0.05$ ; power = 80%). The recruitment was carried out from July 2014 to April 2015. At screening, anthropometric measurements were performed according to the NHLBI practical guidelines (http://www.nhlbi.nih.gov). Eligible participants who provided urine, stool and blood samples were included for analysis. One sample of urine was collected during three consecutive days for the diagnosis of schistosomiasis. One stool sample was collected for assessing STH infection. Venous blood was collected between 8 and 10 am after an overnight-fast using S-Monovette EDTA tube (Sarstetd) for hematological analysis and BD Vacutainer SST II dry tubes (Beckton Dickinson) for serum biochemical analysis. Pregnant women (urinary  $\beta$ -HCG positive test), subjects with a family history of diabetes, non-diagnosed diabetic subjects (fasting serum glucose >7 mM), anyone positive for *Plasmodium falciparum* parasites, and any participants negative for the presence of S. haematobium eggs in urine but infected with at least one other helminths were excluded from the study (S1 Fig). At the end, 71 clinically healthy participants were included and two study groups were considered: a S. haematobium egg-positive  $(Sh^+)$  group with all individuals clinically asymptomatic for schistosomiasis (n = 32), and a helminth negative  $(Sh^{-})$  group including participants negative for schistosomiasis and STH infection (n = 39). Individuals found positive for schistosomiasis (positive for

urine eggs) and/or for STH infections (positive for stool eggs) were later treated with a single dose of 40 mg/kg praziquantel and/or with 400 mg of albendazole daily for three consecutive days, respectively, and those infected with *P. falciparum* were treated with 80/480 mg artemether/lumefantrine tablet twice a day during 3 consecutive days.

#### Parasitological examination

At inclusion, schistosomiasis was evaluated using the microscopic urine filtration method as described [17]. Participants were classified as  $Sh^+$  if at least one egg was detected, and as  $Sh^-$  if three consecutive urine screenings were negative. The infection with STH, including *Ascaris lumbricoides*, *Trichuris trichiura* and *Necator americanus* hookworms, was determined in fresh stool samples using the Kato-Katz technique [17]. A modified agar coproculture protocol was used for the detection of hookworms and *Strongyloides* larva. A thick blood smear was analyzed by microscopic examination for detecting *Plasmodium falciparum* parasites [17].

## Hematological and biochemical analysis

Hematological analysis was performed using an ABX Pentra 60 hematology system (Horiba Medical). For biochemical analysis, serum alanine aminotransferase (ALAT), aspartate aminotransferase (ASAT), glucose, total cholesterol (TC), high density lipoprotein-cholesterol (HDL-c) and triglycerides (TG) were measured on a Modular Analytics P-800 system (Roche Diagnostics). Low density lipoprotein-cholesterol (LDL-c) was calculated applying the Friedewald calculation. Serum insulin and C-peptide were measured on an Immulite 2500 automated system (Siemens Healthcare Diagnostics). High-sensitivity C-reactive protein (hsCRP) was measured using a multiplex array (Meso Scale Discovery). TIgE was measured by ELISA, as previously described [16]. HOMA-IR was calculated as fasting insulin (mU/L) x fasting glucose (mmol/L)/22.5.

#### Circulating anodic antigen assay

Serum Circulating Anodic Antigen (CAA) concentrations were determined by the up-converting phosphor lateral flow (UCP-LF) assay as previously described [19]. Standards with known CAA concentrations were included to generate a calibration curve to calculate individual CAA levels and to validate the threshold of the assay; the maintained lower limit of detection was 10 pg/ml.

#### Serum lipidomics

Lipids were extracted from 100  $\mu$ L of serum by the methyl-tert-butylether method and analyzed using Lipidyzer, a direct infusion-tandem mass spectrometry (DI-MS/MS)-based platform (Sciex, Redwood City, California). Lipid concentrations were expressed as nmol/g of serum.

#### Statistical analysis

The statistical analysis was carried out using SPSS 23.0 software package (SPSS Inc) or Graph-Pad prism 8.3 (GraphPad Software). Quantitative variables were reported as mean and standard deviation (SD) if normally distributed, otherwise they were expressed as median and interquartile range (IQR) and log-transformed for analysis. Unpaired *t*-tests were used to compare study groups. Adjustment for confounding factors was done by multivariate analysis using linear regression, and the differences between groups were expressed as mean and 95% confidence interval (CI). Odd ratios were calculated using multiple logistic regression analysis. One-Way ANOVA with Tukey's post hoc test for multiple comparisons was used to compare the significant differences among the CAA stratified groups. Two-Way ANOVA with Tukey's *post hoc* test for multiple comparisons was used for lipidomic analyses of individual lipid species belonging to the same lipid class. *P* values <0.05 were considered to be statistically significant.

### Results

#### Study population characteristics

The anthropometric characteristics of the study population and the hematological and serum biochemical parameters measured at inclusion are shown in Table 1. Based on urine egg count, 39 participants on a total of 71 individuals (55%) were found positive for *S. haemato-bium* and, among them, 7 were also co-infected with at least one STH species (17.9%). There were no differences in age, gender, BMI, and serum levels of hs-CRP, ALAT and ASAT between the two groups (Table 1), whereas a significant increase in both serum total IgE levels

	S. <i>haematobium</i> negative (n = 32)	S. haematobium positive (n = 39)	P-value	Mean difference adjusted for age, sex, BMI and other helminths (95% CI)	P-value
Age (year) (mean, range)	35.7 (18-63)	34.5 (18-63)	0.68		
Male (%)	43.8	48.7			
BMI (kg/m <sup>2</sup> ) (mean, SD)	<b>26.8</b> (6.9)	<b>25.6</b> (4.5)	0.41		
<i>S. haematobium</i> urine eggs (median, IQR)	<b>0</b> (0–0)	12 (4–63)	<0.001		
Other helminths (%)	0	17.9			
Ascaris lumbricoides (%)	0	2.6			
Trichuris trichiura (%)	0	10.3			
Necator americanus (%)	0	7.7			
Strongyloides (%)	0	2.6			
TIgE (IU/L) (median, IQR)*	<b>6216</b> (1545– 17478)	<b>10476</b> (6570– 19740)	0.034	<b>3226</b> (-1584, 8035)	0.19
Eosinophils (%) (mean, SD)*	<b>10.3</b> (8.2)	<b>18.0</b> (9.6)	0.011	7.1 (0.9, 13.4)	0.025
hs-CRP (mg/L) (median, IQR)	<b>1.91</b> (0.49–4.52)	1.71 (0.68-4.07)	0.78	- <b>0.12</b> (-0.81, 0.56)	0.72
ALAT (GPT, U/L) (mean, SD)	<b>19.2</b> (11.9)	17.1 (9.3)	0.42	- <b>3.3</b> (-8.4, 1.8)	0.20
ASAT (GOT, U/L) (mean, SD)	<b>26.4</b> (10.1)	<b>24.2</b> (7.0)	0.29	<b>-2.9</b> (-7.0, 1.2)	0.16
Glucose (mmol/L) (mean, SD)	<b>4.61</b> (1.06)	<b>4.52</b> (0.67)	0.66	- <b>0.14</b> (-0.57, 0.28)	0.51
Insulin (mU/L) (median, IQR)	<b>4.45</b> (2.74–7.21)	4.54 (2.91-9.62)	0.91	<b>0.06</b> (-0.43, 0.55)	0.81
C-peptide (nmol/L) (median, IQR)	<b>0.41</b> (0.32–0.61)	<b>0.36</b> (0.27–0.62)	0.59	- <b>0.05</b> (-0.38, 0.27)	0.75
HOMA-IR (median, IQR)	<b>0.94</b> (0.52–1.32)	<b>1.00</b> (0.57–1.80)	0.87	<b>0.06</b> (-0.45, 0.58)	0.80
TC (mmol/L) (mean, SD)	<b>4.42</b> (0.84)	<b>4.01</b> (0.81)	0.037	<b>-0.30</b> (-0.68, -0.08)	0.11
HDL-C (mmol/L) (mean, SD)	1.44 (0.40)	<b>1.18</b> (0.31)	0.003	- <b>0.24</b> (-0.43, -0.06)	0.009
LDL-C (mmol/L) (mean, SD)	<b>2.56</b> (0.73)	<b>2.50</b> (0.78)	0.74	- <b>0.04</b> (-0.31, 0.38)	0.84
TG (mmol/L) (mean, SD)	<b>0.93</b> (0.50)	<b>0.72</b> (0.21)	0.031	<b>-0.20</b> (-0.39, -0.03)	0.022

Table 1. Characteristics of the study population.

Normally distributed data are presented as means +/- standard deviation (SD) and non-normally distributed data as median +/- interquartile range (IQR). Adjusted mean difference for TIgE, hs-CRP, Insulin, C-peptide, and HOMA-IR were anti-log transformed.

\*, some values are missing (n = 31 in *Sh*- for TIgE; n = 18 in *Sh*- and n = 30 in *Sh*+ for eosinophils). Abbreviations: BMI: body mass index, TIgE: total immunoglobulin E, hs-CRP: high-sensitivity C-reactive protein, ALAT: alanine aminotransferase, ASAT: aspartate aminotransferase, HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance, TC: total cholesterol, HDL-C: high density lipoprotein-cholesterol, LDL-C: low density lipoprotein cholesterol, TG: triglycerides.

https://doi.org/10.1371/journal.pntd.0008464.t001

(+54%;P = 0.025) and blood eosinophils (+57%;P = 0.028) was observed in *S. haematobium*-infected individuals (*Sh*<sup>+</sup>).

# Effects of *S. haematobium* infection on HOMA-IR, circulating cholesterol and triglyceride levels

To assess the impact of *S. haematobium* infection on glucose homeostasis and insulin resistance, the serum levels of glucose, insulin and C-peptide were measured in *Sh*<sup>-</sup> and *Sh*<sup>+</sup> individuals and the HOMA-IR index calculated. As shown in Table 1, these parameters were not different between the two groups. Next, the effect of *S. haematobium* infection on lipid homeostasis was assessed and serum TC (-9%;P = 0.037), HDL-C (-18%;P = 0.003) and TG (-23%;P = 0.031) levels were found to be lower in *Sh*<sup>+</sup> when compared to *Sh*<sup>-</sup> individuals. After adjustment for age, sex, BMI, and other helminth infection, we observed similar effects, although it remained only significant for HDL-C and TG (Table 1). Multivariate logistic regression analysis was also performed to assess the influence of S. haematobium infection on the abovementioned metabolic parameters and gave similar outcomes (S1 Table). Importantly, the reductions in serum TC, HDL-C and TG levels were mostly seen in overweight/obese infected individuals when the population was stratified according to low/normal (<25) or high (>25) BMI (Table 2). Of note, removing the 7 Sh<sup>+</sup> individuals with other helminth infection from the analyses has only marginal impact on the results and does not affect our conclusions.

# Association between *S. haematobium* infection intensity and serum lipid levels

When measuring serum CAA concentrations, a highly sensitive method for assessing infection intensity [20], we found that a significant fraction of  $Sh^-$  individuals (34%) scored positive for CAA (>10pg/ml), indicating active worm infection, whereas few  $Sh^+$  individuals (9%) returned a CAA concentration below the threshold value of the assay (S2 Fig).

After re-analysis of the data using the serum CAA levels as diagnostic criteria, we found that *S. haematobium* infection was still associated with significant increase in blood eosinophils (+70%;P = 0.040) and reduction in serum TC (-10%;P = 0.020) levels after adjustment for age, sex and BMI (Table 3). Multivariate logistic regression analysis gave similar outcomes (S2 Table). Similar outcomes were found when stratifying the population according to the intensity of the infection using CAA levels, with some non-significant trends for reduced serum levels also observed for HDL-C, LDL-C and TG (S3 Table). When stratifying the population according to BMI, a significant decrease in serum TC levels was observed only in overweight/ obese infected individuals (S4 Table). In line with this, significant negative correlations were found between serum CAA and TC, HDL-C, LDL-C and TG levels, respectively, in both non-stratified whole population and obese/overweight individuals but not in lean subjects (Fig 1).

In order to further investigate the effects of *S. haematobium* infection on circulating lipids and lipoproteins, we performed comprehensive quantitative lipidomic analysis of serum from lean and overweight/obese individuals. The levels of the individual lipid species (S5 Table) belonging to the same lipid class were summed up for an approximate measure of the total serum levels for each lipid class (Fig 2A–2B). We confirmed that total cholesteryl ester (CE) and TG levels were significantly higher in overweight/obese individuals when compared to lean subjects. A significant reduction of CE was observed in both groups of infected subjects but only correlated with the intensity of infection in obese/overweight individuals. Total phosphatidylcholine (PC) and TG levels were also found to be reduced in obese/overweight but not lean infected individuals (Fig 2A–2B). When zooming into the individual lipid species, *S. haematobium* infection was seen to be associated with a significant reduction of 6 PCs (Fig 2C), 3

	BMI <25				BMI >25					
	S. haematobium negative (n = 17)	S. haematobium positive (n = 22)	P-value	Mean difference adjusted for age, sex and other helminths (95% CI)	P- value	S. haematobium negative (n = 15)	S. haematobium positive (n = 17)	P-value	Mean difference adjusted for age, sex and other helminths (95% CI)	P- value
Age (year) (mean, range)	<b>34.3</b> (18–63)	<b>34.5</b> (18–58)	0.96			37.4 (19-49)	<b>34.5</b> (18–63)	0.45		
Male (%)	41.2	50.0				46.7	47.1			
BMI (kg/m <sup>2</sup> ) (mean, SD)	<b>21.6</b> (2.4)	<b>22.2</b> (1.8)	0.40	<b>0.8</b> (-0.7, 2.2)	0.30	<b>32.6</b> (5.6)	<b>30.0</b> (3.0)	0.11	-2.8 (-5.8, 0.2)	0.06
<b>S. haematobium</b> <b>urine eggs</b> (median, IQR)	0	10 (4–62)	<0.001			0	<b>30</b> (5–59)	<0.001		
Other helminths (%)	0	13.6				0	23.5			
Ascaris lumbricoides (%)	0	0				0	5.9			
Trichuris trichiura (%)	0	4.5				0	17.6			
Necator americanus (%)	0	9.1				0	5.9			
Strongyloides (%)	0	4.5				0	0			
<b>TIgE (IU/L)</b> (median, IQR)*	<b>5912</b> (1446– 19547)	<b>10470</b> (5589– 21261)	0.16	<b>2156</b> (-4263, 8576)	0.50	<b>6562</b> (1447– 11527)	<b>10476</b> (6899– 18669)	0.07	<b>3916</b> (-4020, 11852)	0.32
Eosinophils (%) (mean, SD)*	<b>12.0</b> (9.0)	<b>20.3</b> (9.8)	0.08	<b>8.3</b> (-2.0, 18.6)	0.11	<b>9.2</b> (8.0)	15.1 (8.7)	0.12	<b>5.6</b> (-3.1. 14.2)	0.19
hs-CRP (mg/L) (median, IQR)	<b>2.15</b> (0.55– 6.82)	<b>1.19</b> (0.52– 3.42)	0.28	- <b>0.52</b> (-1.50, 0.46)	0.29	<b>1.79</b> (0.46– 3.78)	<b>2.63</b> (1.06– 5.96)	0.28	<b>0.31</b> (-0.62, 1.24)	0.50
ALAT (GPT, U/L) (mean, SD)	<b>17.0</b> (10.6)	<b>16.2</b> (5.9)	0.76	-1.1 (-6.8., 4.6)	0.69	<b>21.6</b> (13.2)	<b>18.3</b> (12.5)	0.47	- <b>6.2</b> (-15.8.9, 3.3)	0.19
ASAT (GOT, U/L) (mean, SD)	<b>25.6</b> (10.5)	<b>23.9</b> (6.1)	0.53	- <b>2.5</b> (-8.0, 3.0)	0.36	27.3 (10.0)	<b>24.7</b> (8.2)	0.42	- <b>3.5</b> (-10.3, 3.2)	0.29
Glucose (mmol/L) (mean, SD)	<b>4.47</b> (1.19)	<b>4.60</b> (0.62)	0.66	<b>0.10</b> (-0.50, 0.70)	0.73	<b>4.76</b> (0.91)	<b>4.41</b> (0.73)	0.23	- <b>0.49</b> (-1.14, 0.15)	0.13
<b>Insulin (mU/L)</b> (median, IQR)	<b>4.55</b> (2.78– 7.93)	<b>5.08</b> (2.91– 9.88)	0.77	<b>0.06</b> (-0.59, 0.71)	0.84	<b>4.22</b> (2.68–7.16)	<b>3.76</b> (2.44– 8.55)	0.87	<b>0.06</b> (-0.75, 0.87)	0.88
C-peptide (nmol/ L) (median, IQR)	<b>0.38</b> (0.31– 0.58)	<b>0.43</b> (0.29– 0.68)	0.93	<b>0.02</b> (-0.44, 0.49)	0.92	<b>0.42</b> (0.32– 1.07)	<b>0.32</b> (0.26– 0.54)	0.36	- <b>0.12</b> (-0.63, 0.39)	0.63
HOMA-IR (median, IQR)	<b>0.93</b> (0.49– 1.33)	<b>1.10</b> (0.59–2.01)	0.58	<b>0.15</b> (-0.54, 0.83)	0.66	<b>0.99</b> (0.52– 1.32)	<b>0.80</b> (0.44– 1.71)	0.73	- <b>0.04</b> (-0.89, 0.81)	0.92
TC (mmol/L) (mean, SD)	4.11 (0.75)	<b>3.98</b> (0.70)	0.60	- <b>0.06</b> (-0.49, 0.36)	0.76	<b>4.78</b> (0.81)	<b>4.04</b> (0.96)	0.025	- <b>0.52</b> (-1.16, 0.12)	0.11
HDL-C (mmol/L) (mean, SD)	1.43 (0.47)	<b>1.21</b> (0.36)	0.12	- <b>0.23</b> (-0.52, 0.07)	0.12	1.46 (0.32)	1.13 (0.22)	0.002	- <b>0.27</b> (-0.48, -0.06)	0.013
LDL-C (mmol/L) (mean, SD)	2.30 (0.54)	2.43 (0.69)	0.52	<b>0.21</b> (-0.16, 0.57)	0.26	2.85 (0.82)	<b>2.59</b> (0.89)	0.39	- <b>0.10</b> (-0.74, 0.53)	0.74
TG (mmol/L) (mean, SD)	0.83 (0.24)	0.74 (0.23)	0.22	- <b>0.10</b> (-0.25, 0.05)	0.18	1.05 (0.69)	0.70 (0.18)	0.07	- <b>0.31</b> (-0.66, 0.05)	0.08

#### Table 2. Characteristics of the study population stratified according to body mass index.

Normally distributed data are presented as means +/- standard deviation (SD) and non-normally distributed data as median +/- interquartile range (IQR). Adjusted mean difference for TIgE, hs-CRP, Insulin, C-peptide and HOMA-IR were anti-log transformed.

\*, some values are missing (for TIgE n = 14 in BMI>25 and *Sh*-; for eosinophils n = 6 in BMI<25 and *Sh*-, n = 17 in BMI<25 and *Sh*+ and n = 9 in BMI>25 and *Sh*-). Abbreviations: BMI: body mass index; TIgE: total immunoglobulin E; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglycerides.

https://doi.org/10.1371/journal.pntd.0008464.t002

	CAA<10 pg/ml (n = 21)	CAA>10 pg/ml (n = 50)	P-value	Mean difference adjusted for age, sex and BMI (95% CI)	P-value
Age (year) (mean, range)	<b>39.2</b> (18–63)	<b>33.3</b> (18–63)	0.07		
Male (%)	42.9	48			
BMI (kg/m <sup>2</sup> ) (mean, SD)	<b>27.0</b> (6.8)	<b>25.7</b> (5.2)	0.40		
TIgE (IU/L) (median, IQR)*	<b>5559</b> (700–17761)	<b>10207</b> (6104–18408)	0.025	4485 (-618, 9588)	0.08
Eosinophils (%) (mean, SD)*	<b>9.9</b> (7.5)	17.3 (9.8)	0.028	<b>6.9</b> (0.3, 13.4)	0.040
hs-CRP (mg/L) (median, IQR)	<b>1.66</b> (0.50–2.69)	2.27 (0.68-5.07)	0.24	<b>0.58</b> (-0.13, 1.29)	0.11
ALAT (GPT, U/L) (mean, SD)	<b>16.3</b> (7.3)	<b>18.8</b> (11.7)	0.38	<b>3.3</b> (-1.6, 8.2)	0.18
ASAT (GOT, U/L) (mean, SD)	<b>23.0</b> (5.3)	<b>26.2</b> (9.5)	0.08	<b>2.43</b> (-1.9, 6.8)	0.20
Glucose (mmol/L) (mean, SD)	4.44 (1.12)	<b>4.61</b> (0.73)	0.44	<b>0.30</b> (-0.15, 0.75)	0.18
Insulin (mU/L) (median, IQR)	4.55 (2.76-7.19)	4.44 (2.89-9.63)	0.94	<b>0.09</b> (-0.43, 0.61)	0.73
C-peptide (nmol/L) (median, IQR)	<b>0.41</b> (0.31–0.69)	0.37 (0.27-0.62)	0.33	<b>-0.12</b> (-0.47, 0.22)	0.48
HOMA-IR (median, IQR)	<b>0.94</b> (0.46–1.22)	<b>0.95</b> (0.57–1.80)	0.71	<b>-0.19</b> (-0.68, 0.29)	0.43
TC (mmol/L) (mean, SD)	4.62 (0.90)	<b>4.01</b> (0.80)	0.005	- <b>0.4</b> 7 (-0.86, -0.07)	0.020
HDL-C (mmol/L) (mean, SD)	1.42 (0.38)	<b>1.24</b> (0.36)	0.06	- <b>0.18</b> (-0.38, 0.02)	0.07
LDL-C (mmol/L) (mean, SD)	2.78 (0.82)	2.42 (0.70)	0.06	<b>-0.24</b> (-0.61, 0.12)	0.19
TG (mmol/L) (mean, SD)	<b>0.91</b> (0.56)	0.78 (0.28)	0.31	- <b>0.09</b> (-0.28, 0.10)	0.34

Table 3. Characteristics of the study population stratified according to CAA levels.

Normally distributed data are presented as means +/- standard deviation (SD) and non-normally distributed data as median +/- interquartile range (IQR). Adjusted mean difference for TIgE, hs-CRP, Insulin, C-peptide and HOMA-IR were anti-log transformed.

\*, some values are missing (n = 49 in CAA>10pg/ml for TIgE; n = 11 in CAA<10pg/ml and n = 34 in CAA>10pg/ml for eosinophils). Abbreviations: BMI: body mass index; TIgE: total immunoglobulin E; CAA: circulating anodic antigen; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglycerides.

https://doi.org/10.1371/journal.pntd.0008464.t003

CEs (Fig 2D) and 6 TGs (Fig 2E), especially among the most abundant species which are also found to be increased in overweight/obese individuals (*e.g.* PC(16:0/18:1), CE (18:2) and TG (52:2))[21, 22].

#### Discussion

The main objective of the present study was to investigate the effects of *S. haematobium* infection on metabolic outcomes, especially on whole-body lipid/lipoprotein homeostasis, in lean and overweight/obese individuals residing in a highly endemic rural area. Although no effect of *S. haematobium* infection was observed on HOMA-IR, whatever the BMI status, a strong reduction in serum TC levels was consistently observed in infected individuals. Furthermore, negative correlations were also observed between the intensity of active *S. haematobium* infection and the circulating levels of TC, HDL-c LDL-c, TG, and some lipid species associated with cholesterol-rich lipoprotein particles in overweight/obese individuals. Altogether, our findings showed for the first time that *S. haematobium* infection is associated with improved circulating lipid profiles in humans, especially in overweight/obese individuals.

The lowering effect of *S. haematobium* infection on serum TC and cholesterol levels observed in our study is in line with the few cross-sectional investigations conducted so far in countries endemic for other helminth species. Wiria *et al.* have indeed shown that infection with the soil-transmitted helminths *Necator americanus*, *Ascaris lumbricoides* and *Trichuris trichiura* was associated with reduced serum levels of TC and HDL-c in lean Indonesian individuals from Flores Island [23]. Duan *et al.* have also recently reported that serum TG, TC and



**Fig 1.** Associations between intensity of *S. haematobium* infection assessed by circulating anodic antigen levels and serum lipid parameters. The correlations between serum CAA and TC, HDL-C, LDL-C and TG were shown for the whole population ([n = 71], a) and for lean (BMI<25 [n = 39], b) and overweight/obese (BMI>25 [n = 32], c) individuals. *Sh. Schistosoma haematobium*; CAA, Circulating Anodic Antigen; TC, Total Cholesterol; HDL-C, High-Density Lipoprotein-Cholesterol; LDL-C, Low-Density Lipoprotein-Cholesterol; TG, Triglycerides.

https://doi.org/10.1371/journal.pntd.0008464.g001

LDL-c levels were significantly lower in lean Chinese subjects with chronic *S. japonicum* infection when compared to uninfected individuals [24]. Finally, during the preparation of the manuscript, the results of a study investigating the effects of *S. mansoni* infection on metabolic outcomes in the framework of the LaVIISWA trial conducted in the highly-endemic region of Lake Victoria in Uganda became available [25]. Interestingly, a negative association between the intensity of *S. mansoni* infection and serum TC, HDL-c, LDL-c and TG levels was found [25]. Moreover, community-wide intensive anthelminthic treatment that resulted in lower *S. mansoni* prevalence was also associated with a trend towards increased TC and LDL-c, suggesting that deworming might revert some of the effects induced by helminth infection on lipid profile [25]. However, it is worth to mentioning that our study is the first one



**Fig 2. Effects of** *S. haematobium* infection on serum lipid profiles in lean and obese subjects. Lipidomics analysis was performed using the Lipidyzer platform on serum from lean (BMI<25, a) and overweight/obese (BMI>25, b) individuals, and the data for the lipid classes (a) and for the various PC (c), CE (d) and TG (e) species were stratified in each population according to 'below detection threshold' (<10pg/ml), 'medium/high' (10<CAA<3000pg/ml) or 'high' (>3000pg/ml) CAA levels. PC, Phosphatidylcholine; PE, Phosphatidylethanolamine; LPC, Lysophosphatidylcholine; LPE, Lysophosphatidylethanolamine; SM, Sphingomyelin; CE, Cholesterylester; DG, Diglycerides; TG, Triglycerides; FFA, Free-fatty acids. \*, p<0.05 versus CAA < 10; #, p<0.05 versus BMI <25.

https://doi.org/10.1371/journal.pntd.0008464.g002

investigating the impact of helminth infection in overweight/obese individuals, who have elevated lipid/cholesterol levels and higher risk for developing cardiometabolic diseases.

Several underlying mechanisms may account for the effects of helminth on lipid homeostasis, involving 1) parasitism of host dietary nutrients through hijacking of lipid resources for their own survival and reproduction, 2) alteration of gut microbiota and intestinal lipid metabolism, and 3) immune-dependent or -independent regulation of lipid metabolism in immune cells and/or peripheral tissues by helminths and their molecules. Although adult schistosomes cannot oxidize exogenous FA [26], they can use and modify host lipid for biosynthetic purposes by scavenging cholesterol and incorporating FA as phospholipids and neutral lipids in the worm eggs, notably for membrane formation of the developing miracidium [27]. It is therefore tempting to speculate that part of the reduction in some lipoprotein and TG species observed in *S. haematobium*-infected individuals might result from their active transfer to the eggs produced by female worms and their subsequent elimination in the urine. Further studies are however required to investigate this point.

During the last decade, the gut microbiota has emerged as an important player in host metabolic homeostasis, notably by regulating intestinal lipid processing from dietary fat and nutrient metabolism in peripheral organs, but also in shaping the local and systemic immune response [28, 29]. Gut microbiota, via local production of various metabolites, can impact host lipid metabolism by modulating bile acid metabolism and/or a gut-brain axis involved in the control of whole-body lipid homeostasis [30, 31]. Since various helminth species, including *S. haematobium* [32], have been reported to affect gut microbiota in humans [33], one may hypothesize that this could underlie some of the reduction in TC and TG levels observed in helminth infected individuals, secondary to alterations of intestinal lipid absorption, hepatic cholesterol and BA biosynthesis, secretion and reuptake, or hydrolysis of TG-rich lipoproteins by peripheral metabolic tissues. Further studies would be required for investigating the putative associations between helminth-induced changes in specific gut bacterial species and serum/fecal lipid/BA profiles.

Infection with schistosomes is associated with eosinophilia, parasite-specific IgE and elevated type-2 cytokines, which are all hallmarks of a  $T_{H}^{2}$  immune response [10, 34]. One may speculate that some of the beneficial effects observed on serum lipid profile from obese/overweight individuals could result from this helminth-induced immunomodulation through direct or indirect on metabolic organs. The liver plays an essential role in lipid/cholesterol metabolism, notably by regulating de novo lipogenesis, cholesterol biosynthesis and VLDL-TG secretion, and facilitating the clearance of circulating HDL-c via selective uptake through membrane scavenger receptors. Interestingly, type-2 cytokines, such as IL-4 and IL-13, have been shown to regulate hepatic metabolism in rodents through direct interaction with their cognate receptors expressed on hepatocytes [35, 36]. This type-2 micro-environment could also have some impact on lipid metabolism in other peripheral organs, notably in AT through modulation of TG lipolysis/lipogenesis. It is also worth mentioning that some of the S. haematobium worm- and/or egg-derived molecules released inside the host may also contribute to tissue-specific modulation of lipid metabolism through immune-independent direct interaction with metabolic cells [11, 37]. Furthermore, intracellular nutrient metabolism has been shown to be central in the regulation of immune cell activation, differentiation and function [38, 39]. As such, one may speculate that the decrease in serum lipid levels observed in S. haematobium-infected individuals may also partly result from increased FA and cholesterol metabolism by various circulating and/or tissue-resident immune cells secondary to helminthinduced host immunomodulation, as reported for example in hepatic macrophages from infected mice [40].

The main limitations of our study are based on its observational nature, which prevents firm conclusions on causal relationship, and the relatively small sample size, especially after stratification according to BMI or CAA levels. Future studies involving larger cohort of lean and obese individuals and/or deworming intervention using anti-helminthic drug in such population would be required. In contrast to what has been reported with STH [12, 13, 16], we did not find any association between *S. haematobium* infection and improved whole-body insulin resistance as assessed by HOMA-IR. This discrepancy could be explained by differences in helminth species but also eventually in age and/or lifestyle of the population studied,

notably their nutritional habits. Furthermore, past infections with *S. haematobium* could have already imprinted their effect on insulin sensitivity, masking the influence of current infection. In this regard, trials where anthelminthic treatment is used or controlled human infection in naïve individuals with various helminth species [41-43] can better answer this question.

# **Supporting information**

**S1 Fig. Flow chart of study participants enrolment.** Sh, *Schistosoma haematobium*; STH, Soil-transmitted helminths; BMI, Body Mass Index. (TIF)

**S2 Fig. Serum CAA levels in urine egg negative and positive individuals.** The serum CAA concentrations were measured in Sh- and Sh+ individuals diagnosed by presence of urine eggs (a), and the data were stratified in each population according to 'below detection threshold' (<10pg/ml), 'medium/high' (10<CAA<3000pg/ml) or 'high' (>3000pg/ml) CAA levels (b). The correlation between serum CAA levels and urine *S. haematobium* egg counts (c) in the whole population was plotted. (TIF)

**S1 Table. Multiple logistic regression analysis on effect of** *Schistosoma haematobium* infection diagnosed by urine egg microscopy on various metabolic parameters. Abbreviations: CI: confidence interval; OR: Odd ratio; TIgE: total immunoglobulin E; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglyceride.

(DOCX)

**S2 Table. Multiple logistic regression analysis on effect of** *Schistosoma haematobium* infection diagnosed by serum CAA levels on various metabolic parameters. Abbreviations: CI: confidence interval; OR: Odd ratio; TIgE: total immunoglobulin E; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglyceride.

(DOCX)

**S3 Table. Characteristics of the study population stratified according to CAA range.** Normally distributed data are presented as means +/- standard deviation (SD) and non-normally distributed data as median +/- interquartile range (IQR). \*, some values are missing (for TIgE n = 36 in 10<CAA<3000pg/ml; for eosinophils n = 11 in CAA<10pg/ml, n = 24 in 10<CAA<3000pg/ml and n = 10 in CAA>3000pg/ml). Abbreviations: BMI: body mass index; TIgE: total immunoglobulin E; CAA: circulating anodic antigen; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglyceride

(DOCX)

**S4** Table. Characteristics of the study population stratified according to CAA levels and body mass index. Normally distributed data are presented as means +/- standard deviation (SD) and non-normally distributed data as median +/- interquartile range (IQR). Adjusted

mean difference for TIgE, hs-CRP, Insulin, C-peptide and HOMA-IR were anti-log transformed. \*, some values are missing (for TIgE n = 21 in BMI>25 and CAA>10pg/ml, for eosinophils n = 4 in BMI<25 and CAA<10pg/ml, n = 19 in BMI<25 and CAA>10pg/ml, n = 7 in BMI>25 and CAA<10pg/ml and n = 15 in BMI>25 and CAA>10pg/ml). Abbreviations: BMI: body mass index; TIgE: total immunoglobulin E; CAA: circulating anodic antigen; hs-CRP: high-sensitivity C-reactive protein; ALAT: alanine aminotransferase; ASAT: aspartate aminotransferase; HOMA-IR: HOmeostatic Model Assessment for Insulin Resistance; TC: total cholesterol; HDL-C: high density lipoprotein-cholesterol; LDL-C: low density lipoprotein cholesterol; TG: triglycerides.

(DOCX)

**S5 Table. Serum lipidomics in subjects stratified according to CAA levels.** Data are presented as means (+/- SD). Abbreviations: CAA: circulating anodic antigen; PC: Phosphatidylcholine; PE: Phosphatidylethanolamine; LPC: Lysophosphatidylcholine; LPE: Lysophosphatidylethanolamine; SM: Sphingomyelin; CE: Cholesterylester; DG: Diglycerides; TG: Triglycerides; FFA: Free-fatty acids. (DOCX)

## Acknowledgments

The authors want to thank the participants who took part in this study.

## **Author Contributions**

Conceptualization: Bruno Guigas.

Data curation: Jeannot F. Zinsou, Bruno Guigas.

Formal analysis: Jeannot F. Zinsou, Jacqueline J. Janse, Bruno Guigas.

Funding acquisition: Martin Giera, Maria Yazdanbakhsh, Bruno Guigas.

- Investigation: Jeannot F. Zinsou, Yabo Y. Honpkehedji, Jean Claude Dejon-Agobé, Noemí García-Tardón, Pytsje T. Hoekstra, Marguerite Massinga-Loembe, Martin Giera.
- Methodology: Jeannot F. Zinsou, Noemí García-Tardón, Pytsje T. Hoekstra, Paul L. A. M. Corstjens, Govert J. van Dam, Martin Giera, Bruno Guigas.
- **Project administration:** Jeannot F. Zinsou, Marguerite Massinga-Loembe, Ayola A. Adegnika, Bruno Guigas.

Supervision: Peter G. Kremsner, Maria Yazdanbakhsh, Ayola A. Adegnika, Bruno Guigas.

Visualization: Bruno Guigas.

Writing - original draft: Jeannot F. Zinsou, Bruno Guigas.

Writing – review & editing: Jean Claude Dejon-Agobé, Pytsje T. Hoekstra, Paul L. A. M. Corstjens, Martin Giera, Maria Yazdanbakhsh, Ayola A. Adegnika, Bruno Guigas.

#### References

- Hotez PJ, Alvarado M, Basanez MG, Bolliger I, Bourne R, Boussinesq M, et al. The global burden of disease study 2010: interpretation and implications for the neglected tropical diseases. PLoS Negl Trop Dis. 2014; 8(7):e2865. Epub 2014/07/25. https://doi.org/10.1371/journal.pntd.0002865 PMID: 25058013; PubMed Central PMCID: PMC4109880.
- 2. Karagiannis-Voules DA, Biedermann P, Ekpo UF, Garba A, Langer E, Mathieu E, et al. Spatial and temporal distribution of soil-transmitted helminth infection in sub-Saharan Africa: a systematic review and

geostatistical meta-analysis. Lancet Infect Dis. 2015; 15(1):74–84. Epub 2014/12/10. https://doi.org/10. 1016/S1473-3099(14)71004-7 PMID: 25486852.

- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. Lancet. 2014; 383 (9936):2253–64. Epub 2014/04/05. https://doi.org/10.1016/S0140-6736(13)61949-2 PMID: 24698483; PubMed Central PMCID: PMC4672382.
- McManus DP, Dunne DW, Sacko M, Utzinger J, Vennervald BJ, Zhou XN. Schistosomiasis. Nat Rev Dis Primers. 2018; 4(1):13. Epub 2018/08/11. https://doi.org/10.1038/s41572-018-0013-8 PMID: 30093684.
- 5. IDF. IDF Diabetes Atlas. 2017;8th edn:http://www.diabetesatlas.org/.
- WHO. Global status report on noncommunicable diseases 2014. <u>http://www.who.int/nmh/publications/ncd-status-report-2014/en/.</u>
- van Crevel R, van de Vijver S, Moore DA. The global diabetes epidemic: what does it mean for infectious diseases in tropical countries? Lancet Diabetes Endocrinol. 2016. <u>https://doi.org/10.1016/S2213-8587(16)30081-X PMID: 27499355</u>.
- Hotamisligil GS. Inflammation, metaflammation and immunometabolic disorders. Nature. 2017; 542 (7640):177–85. https://doi.org/10.1038/nature21363 PMID: 28179656.
- Lackey DE, Olefsky JM. Regulation of metabolism by the innate immune system. Nat Rev Endocrinol. 2016; 12(1):15–28. Epub 2015/11/11. https://doi.org/10.1038/nrendo.2015.189 PMID: 26553134.
- Maizels RM, McSorley HJ. Regulation of the host immune system by helminth parasites. J Allergy Clin Immunol. 2016; 138(3):666–75. Epub 2016/08/02. https://doi.org/10.1016/j.jaci.2016.07.007 PMID: 27476889; PubMed Central PMCID: PMC5010150.
- van der Zande HJP, Zawistowska-Deniziak A, Guigas B. Immune Regulation of Metabolic Homeostasis by Helminths and Their Molecules. Trends Parasitol. 2019; 35(10):795–808. Epub 2019/09/08. https:// doi.org/10.1016/j.pt.2019.07.014 PMID: 31492623.
- Wiria AE, Hamid F, Wammes LJ, Prasetyani MA, Dekkers OM, May L, et al. Infection with Soil-Transmitted Helminths Is Associated with Increased Insulin Sensitivity. PLoS One. 2015; 10(6):e0127746. https://doi.org/10.1371/journal.pone.0127746 PMID: 26061042; PubMed Central PMCID: PMC4464734.
- Chen Y, Lu J, Huang Y, Wang T, Xu Y, Xu M, et al. Association of previous schistosome infection with diabetes and metabolic syndrome: a cross-sectional study in rural China. J Clin Endocrinol Metab. 2013; 98(2):E283–7. Epub 2013/01/01. https://doi.org/10.1210/jc.2012-2517 PMID: 23275524.
- Hays R, Esterman A, Giacomin P, Loukas A, McDermott R. Does Strongyloides stercoralis infection protect against type 2 diabetes in humans? Evidence from Australian Aboriginal adults. Diabetes Res Clin Pract. 2015; 107(3):355–61. Epub 2015/02/07. https://doi.org/10.1016/j.diabres.2015.01.012 PMID: 25656764.
- Rajamanickam A, Munisankar S, Bhootra Y, Dolla C, Thiruvengadam K, Nutman TB, et al. Metabolic consequences of concomitant Strongyloides stercoralis infection in Type 2 diabetes mellitus. Clin Infect Dis. 2018. Epub 2018/11/09. https://doi.org/10.1093/cid/ciy935 PMID: 30407548.
- Tahapary DL, de Ruiter K, Martin I, Brienen EAT, van Lieshout L, Cobbaert CM, et al. Effect of Anthelmintic Treatment on Insulin Resistance: A Cluster-Randomized, Placebo-Controlled Trial in Indonesia. Clin Infect Dis. 2017; 65(5):764–71. Epub 2017/05/05. <u>https://doi.org/10.1093/cid/cix416</u> PMID: 28472383.
- Dejon-Agobe JC, Edoa JR, Honkpehedji YJ, Zinsou JF, Adegbite BR, Ngwese MM, et al. Schistosoma haematobium infection morbidity, praziquantel effectiveness and reinfection rate among children and young adults in Gabon. Parasit Vectors. 2019; 12(1):577. Epub 2019/12/12. https://doi.org/10.1186/ s13071-019-3836-6 PMID: 31823806; PubMed Central PMCID: PMC6905022.
- Dejon-Agobe JC, Zinsou JF, Honkpehedji YJ, Ateba-Ngoa U, Edoa JR, Adegbite BR, et al. Schistosoma haematobium effects on Plasmodium falciparum infection modified by soil-transmitted helminths in school-age children living in rural areas of Gabon. PLoS Negl Trop Dis. 2018; 12(8):e0006663. Epub 2018/08/07. https://doi.org/10.1371/journal.pntd.0006663 PMID: 30080853; PubMed Central PMCID: PMC6095623.
- Corstjens PL, De Dood CJ, Kornelis D, Fat EM, Wilson RA, Kariuki TM, et al. Tools for diagnosis, monitoring and screening of Schistosoma infections utilizing lateral-flow based assays and upconverting phosphor labels. Parasitology. 2014; 141(14):1841–55. Epub 2014/06/17. https://doi.org/10.1017/S0031182014000626 PMID: 24932595; PubMed Central PMCID: PMC4265670.
- de Dood CJ, Hoekstra PT, Mngara J, Kalluvya SE, van Dam GJ, Downs JA, et al. Refining Diagnosis of Schistosoma haematobium Infections: Antigen and Antibody Detection in Urine. Front Immunol. 2018; 9:2635. Epub 2018/11/30. <u>https://doi.org/10.3389/fimmu.2018.02635</u> PMID: <u>30487796</u>; PubMed Central PMCID: PMC6246739.

- Yin X, Willinger CM, Keefe J, Liu J, Fernandez-Ortiz A, Ibanez B, et al. Lipidomic profiling identifies signatures of metabolic risk. EBioMedicine. 2020; 51:102520. Epub 2019/12/27. https://doi.org/10.1016/j.ebiom.2019.10.046 PMID: 31877415; PubMed Central PMCID: PMC6938899.
- 22. Tonks KT, Coster AC, Christopher MJ, Chaudhuri R, Xu A, Gagnon-Bartsch J, et al. Skeletal muscle and plasma lipidomic signatures of insulin resistance and overweight/obesity in humans. Obesity (Silver Spring). 2016; 24(4):908–16. Epub 2016/02/27. https://doi.org/10.1002/oby.21448 PMID: 26916476; PubMed Central PMCID: PMC6585800.
- Wiria AE, Wammes LJ, Hamid F, Dekkers OM, Prasetyani MA, May L, et al. Relationship between carotid intima media thickness and helminth infections on Flores Island, Indonesia. PLoS One. 2013; 8 (1):e54855. Epub 2013/02/01. https://doi.org/10.1371/journal.pone.0054855 PMID: 23365679; PubMed Central PMCID: PMC3554693.
- Duan Q, Xiong L, Liao C, Liu Z, Xiao Y, Huang R, et al. Population based and animal study on the effects of Schistosoma japonicum infection in the regulation of host glucose homeostasis. Acta Trop. 2018; 180:33–41. Epub 2018/01/09. https://doi.org/10.1016/j.actatropica.2018.01.002 PMID: 29309743.
- 25. Sanya RE, Webb EL, Zziwa C, Kizindo R, Sewankambo M, Tumusiime J, et al. The effect of helminth infections and their treatment on metabolic outcomes: results of a cluster-randomised trial. Clin Infect Dis. 2019. Epub 2019/09/11. https://doi.org/10.1093/cid/ciz859 PMID: 31504336.
- Bexkens ML, Mebius MM, Houweling M, Brouwers JF, Tielens AGM, van Hellemond JJ. Schistosoma mansoni does not and cannot oxidise fatty acids, but these are used for biosynthetic purposes instead. Int J Parasitol. 2019; 49(8):647–56. Epub 2019/06/07. https://doi.org/10.1016/j.ijpara.2019.03.005 PMID: 31170410.
- Brouwers JF, Smeenk IM, van Golde LM, Tielens AG. The incorporation, modification and turnover of fatty acids in adult Schistosoma mansoni. Mol Biochem Parasitol. 1997; 88(1–2):175–85. Epub 1997/ 09/01. https://doi.org/10.1016/s0166-6851(97)00091-1 PMID: 9274878.
- Wang Z, Koonen D, Hofker M, Fu J. Gut microbiome and lipid metabolism: from associations to mechanisms. Curr Opin Lipidol. 2016; 27(3):216–24. Epub 2016/04/08. <u>https://doi.org/10.1097/MOL.000000000000308</u> PMID: 27054442.
- Zhang Z, Tang H, Chen P, Xie H, Tao Y. Demystifying the manipulation of host immunity, metabolism, and extraintestinal tumors by the gut microbiome. Signal Transduct Target Ther. 2019; 4:41. Epub 2019/10/23. https://doi.org/10.1038/s41392-019-0074-5 PMID: <u>31637019</u>; PubMed Central PMCID: PMC6799818.
- Wahlstrom A, Sayin SI, Marschall HU, Backhed F. Intestinal Crosstalk between Bile Acids and Microbiota and Its Impact on Host Metabolism. Cell Metab. 2016; 24(1):41–50. Epub 2016/06/21. https://doi. org/10.1016/j.cmet.2016.05.005 PMID: 27320064.
- Kriaa A, Bourgin M, Potiron A, Mkaouar H, Jablaoui A, Gerard P, et al. Microbial impact on cholesterol and bile acid metabolism: current status and future prospects. J Lipid Res. 2019; 60(2):323–32. Epub 2018/11/30. https://doi.org/10.1194/jlr.R088989 PMID: <u>30487175</u>; PubMed Central PMCID: PMC6358303.
- Kay GL, Millard A, Sergeant MJ, Midzi N, Gwisai R, Mduluza T, et al. Differences in the Faecal Microbiome in Schistosoma haematobium Infected Children vs. Uninfected Children. PLoS Negl Trop Dis. 2015; 9(6):e0003861. Epub 2015/06/27. https://doi.org/10.1371/journal.pntd.0003861 PMID: 26114287; PubMed Central PMCID: PMC4482744.
- Rapin A, Harris NL. Helminth-Bacterial Interactions: Cause and Consequence. Trends Immunol. 2018; 39(9):724–33. Epub 2018/06/27. https://doi.org/10.1016/j.it.2018.06.002 PMID: 29941203.
- Colley DG, Secor WE. Immunology of human schistosomiasis. Parasite Immunol. 2014; 36(8):347–57. Epub 2014/08/22. <u>https://doi.org/10.1111/pim.12087</u> PMID: 25142505; PubMed Central PMCID: PMC4278558.
- Ricardo-Gonzalez RR, Red Eagle A, Odegaard JI, Jouihan H, Morel CR, Heredia JE, et al. IL-4/STAT6 immune axis regulates peripheral nutrient metabolism and insulin sensitivity. Proc Natl Acad Sci U S A. 2010; 107(52):22617–22. Epub 2010/12/15. https://doi.org/10.1073/pnas.1009152108 PMID: 21149710; PubMed Central PMCID: PMC3012500.
- Stanya KJ, Jacobi D, Liu S, Bhargava P, Dai L, Gangl MR, et al. Direct control of hepatic glucose production by interleukin-13 in mice. J Clin Invest. 2013; 123(1):261–71. Epub 2012/12/22. https://doi.org/ 10.1172/JCI64941 PMID: 23257358; PubMed Central PMCID: PMC3533296.
- Bhargava P, Li C, Stanya KJ, Jacobi D, Dai L, Liu S, et al. Immunomodulatory glycan LNFPIII alleviates hepatosteatosis and insulin resistance through direct and indirect control of metabolic pathways. Nat Med. 2012; 18(11):1665–72. <u>https://doi.org/10.1038/nm.2962</u> PMID: <u>23104131</u>; PubMed Central PMCID: PMC3493877.

- Hubler MJ, Kennedy AJ. Role of lipids in the metabolism and activation of immune cells. J Nutr Biochem. 2016; 34:1–7. Epub 2016/07/18. https://doi.org/10.1016/j.jnutbio.2015.11.002 PMID: 27424223; PubMed Central PMCID: PMC5694687.
- Tall AR, Yvan-Charvet L. Cholesterol, inflammation and innate immunity. Nat Rev Immunol. 2015; 15 (2):104–16. Epub 2015/01/24. <u>https://doi.org/10.1038/nri3793</u> PMID: <u>25614320</u>; PubMed Central PMCID: PMC4669071.
- 40. Cortes-Selva D, Elvington AF, Ready A, Rajwa B, Pearce EJ, Randolph GJ, et al. Schistosoma mansoni Infection-Induced Transcriptional Changes in Hepatic Macrophage Metabolism Correlate With an Athero-Protective Phenotype. Front Immunol. 2018; 9:2580. Epub 2018/11/30. https://doi.org/10.3389/ fimmu.2018.02580 PMID: 30483256; PubMed Central PMCID: PMC6240656.
- Loukas A, Hotez PJ, Diemert D, Yazdanbakhsh M, McCarthy JS, Correa-Oliveira R, et al. Hookworm infection. Nat Rev Dis Primers. 2016; 2:16088. Epub 2016/12/09. https://doi.org/10.1038/nrdp.2016.88 PMID: 27929101.
- 42. Pierce D, Merone L, Lewis C, Rahman T, Croese J, Loukas A, et al. Safety and tolerability of experimental hookworm infection in humans with metabolic disease: study protocol for a phase 1b randomised controlled clinical trial. BMC Endocr Disord. 2019; 19(1):136. Epub 2019/12/13. <u>https://doi.org/10.1186/ s12902-019-0461-5 PMID: 31829172</u>.
- Roestenberg M, Hoogerwerf MA, Ferreira DM, Mordmuller B, Yazdanbakhsh M. Experimental infection of human volunteers. Lancet Infect Dis. 2018; 18(10):e312–e22. Epub 2018/06/13. https://doi.org/10. 1016/S1473-3099(18)30177-4 PMID: 29891332.